Author's response to reviews

Title: Right coronary artery originating from left anterior descending artery: a case report

Authors:

Hilmi Tokmakoglu (h.tokmakoglu@isnet.net.tr)
Orhan Bozoglan (orhanbozoglan1975@hotmail.com)
Levent Ozdemir (levent.ozdemir@bozok.edu.tr)

Version: 4 Date: 5 June 2010

Author's response to reviews: see over
Dear Editor,

1. Competing interests is added to the text.

2. Reference 14 was changed as reference 17 and also references 15-17 were changed as references 14-16 and corrected references were cited.

3. Figures 3 and 4 are cited within the main manuscript text and all figures were titled individually.

4. Manuscript was checked for any typographical errors.

Yours sincerely